Author's response to reviews

Title: Use of Imaging Biomarkers to Assess Perfusion and Glucose Metabolism in the Skeletal Muscle of Dystrophic Mice

Authors:

Nabeel Ahmad (naahmad3@gmail.com)
Ian Welch (iwelch@uwo.ca)
Robert Grange (rgrange@vt.edu)
Jennifer Hadway (jhadway@lawsonimaging.ca)
Savita Dhanvantari (sdhanvan@lawsonimaging.ca)
David Hill (david.hill@lri.sjhc.london.on.ca)
Ting-Yim Lee (tlee@lawsonimaging.ca)
Lisa M Hoffman (lhoffman@lawsonimaging.ca)

Version: 4 Date: 25 May 2011

Author's response to reviews:

As per your request, we have made the following changes to the format of our manuscript, so that the published article conforms to journal style.

Formatting changes made:

-----------------------------

Major revisions

Figures: Figure legend 5 contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square"). Please ensure the corrected legend does not exceed 250 words.

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). You have Lisa M. Hoffman in the submissions system but Lisa Hoffman in the manuscript, please rectify.

Please cite the appendix in the text or remove from the manuscript: it is currently not cited

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes
above)

Abbreviations: Please format these in to a paragraph in alphabetical order.

Please place the sources of funding in the acknowledgments section.

Please remove the author list from the authors contributions section.

Please move the appendix section to after your conclusions and entitle 'appendix'

Please delete the authors information section.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Please place the table title above the table

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.